

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Reuter, Wolf Heinrich; Tkačevs, Oļegs; Vilerts, Kārlis

Working Paper On the design of stabilising fiscal rules

Arbeitspapier, No. 11/2018

Provided in Cooperation with: German Council of Economic Experts

Suggested Citation: Reuter, Wolf Heinrich; Tkačevs, Oļegs; Vilerts, Kārlis (2018) : On the design of stabilising fiscal rules, Arbeitspapier, No. 11/2018, Sachverständigenrat zur Begutachtung der Gesamtwirtschaftlichen Entwicklung, Wiesbaden

This Version is available at: https://hdl.handle.net/10419/193674

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU



On the design of stabilising fiscal rules

Wolf Heinrich Reuter (German Council of Economic Experts)

> Oļegs Tkačevs Kārlis Vilerts (both Bank of Latvia)

Working Paper 11/2018*) December, 2018

*) Working papers reflect the personal views of the authors and not necessarily those of the German Council of Economic Experts.

ON THE DESIGN OF STABILISING FISCAL RULES

WOLF HEINRICH REUTER OĻEGS TKAČEVS KĀRLIS VILERTS

ABSTRACT

Utilising data of the EU28 Member States for the period 1996–2015, this paper confirms the findings of previous studies that the stipulation of fiscal rules reduces fiscal volatility and consequently contributes to macroeconomic stability. Yet, we document that this result only holds for rules which are designed to be unaffected by the current state of the business cycle, i.e. which are "a-cyclical". Those can, e.g. be budget balance rules that set ceilings in cyclically adjusted terms or expenditure rules that set a limit relative to potential instead of current output. Furthermore, the stringency of fiscal rules amplifies their stabilising effect. Actual year-to-year compliance with fiscal rules seems to play no systematic role, such that effects of the rules can be observed even if they are not complied with year-to-year. Overall, our paper suggests that strong, properly designed numerical rules act as an anchor for fiscal policy makers and contribute to more stable discretionary fiscal policy.

Keywords: fiscal rules, fiscal policy volatility, panel data, compliance

JEL codes: C23, E62, E32, H60

The opinions expressed in this paper are those of the authors – Wolf Heinrich Reuter (German Council of Economic Experts), Olegs Tkačevs (Latvijas Banka) and Kārlis Vilerts (Latvijas Banka) – and do not necessarily reflect the official viewpoint of the German Council of Economic Experts or Latvijas Banka. The authors are grateful to Christoph M. Schmidt, Mārtiņš Bitāns, Gundars Dāvidsons, Konstantīns Beņkovskis as well as participants of internal seminars held at Latvijas Banka, the Latvian Fiscal Discipline Council, Inaugural Baltic Economic Conference in Vilnius and Banka Slovenije public finance workshop in Ljubljana for their very valuable comments and suggestions. Comments by the discussants of our paper Gonzalo Caprirolo (Ministry of Finance, Slovenia) and Marcus Jüppner (Goethe University Frankfurt, Deutsche Bundesbank, Germany) are gratefully acknowledged.

1. INTRODUCTION

The recent sovereign debt crisis was triggered by concerns about the sustainability of public debt in a number of euro area Member States. Supranational fiscal rules in the euro area (laid out in the SGP) appeared insufficient to ensure sustainable fiscal policies before the crisis. This reinforced calls for imposing stricter political constraints on fiscal policy. In 2012, EU Member States agreed on the Fiscal Compact.¹ As a result, many EU countries introduced more stringent numerical fiscal rules and incorporated them into their statutory legislation, including, in some cases, through constitutional amendments. While only seven EU countries had a budget balance, expenditure or debt rule in place in the mid-1990s, this number rose to 25 by 2015. This surge has also been accompanied by a considerable increase in the strength of the rules and fiscal frameworks (IMF (2017)).

The stipulation of numerical constraints on fiscal policy (numerical fiscal rules) usually aims at counteracting the widely recognised deficit bias of governments and limiting the accumulation of a large public debt.² While the positive effect of stipulating fiscal rules on the fiscal discipline has been well documented in empirical studies³, the stabilising effects are not straightforward. On the one hand, explicit numerical restrictions should reduce the uncertainty and volatility of the conduct of discretionary fiscal policy. If this restricts ill-timed or erratic discretionary fiscal policy, which might exacerbate swings in output growth, it increases macroeconomic stability. On the other hand, rules may limit the capacity to use discretionary fiscal policy tools for smoothing out business cycle fluctuations and therefore increase instability. Which of these two effects prevails is *a priori* unclear.

Overall, there already exists some evidence that fiscal constraints dampen the amplitude of business cycles through their stabilising effect on fiscal policy. However, to the best of our knowledge, there has been no discussion of whether these effects depend on certain properties of fiscal rules or actual compliance with them. If the effects of rules depend on their ability to constrain fiscal policy and keep it within the limits set out by them, the rules that are actually complied with year-to-year should be associated with a stronger effect. However, the rules can also act as kind of a benchmark or anchor for fiscal policy which reduces uncertainty and prevents long-lasting deviations from the limits set by the rules. In that case the rules might still have an effect even if they are not complied with year-to-year, and other features of the rules might come into focus. We intend to fill this gap in the literature with this paper by investigating whether and how volatility of fiscal policy is affected by existence and design of fiscal rules and compliance with them.

Our study is related to several streams of literature that analyse the links between fiscal policy volatility, macroeconomic stability, national fiscal rules and compliance with them. Two papers, which we follow methodologically, are the seminal study by Fatas and Mihov (2006) and a more recent one by Badinger and Reuter (2017). The former study demonstrates that fiscal constraints reduce the use of fiscal discretion leading to less volatile output growth. The study by Badinger and Reuter (2017) also confirms the dampening effect of fiscal rules on fiscal policy volatility, particularly if rules are embedded in a more stringent fiscal framework, e.g. by strong legislative support, broader coverage or formal enforcement procedures. However, our paper goes beyond the existing literature by i) employing panel data and thus taking within-country time variation into account (similar to Sacchi and Salotti (2015)), ii) taking design properties of fiscal rules into account, and iii)

¹ The Treaty on Stability, Coordination and Governance in the Economic and Monetary Union.

² For the definition of national fiscal rules, their rationale, a brief historical overview of the earliest rules as well as basic properties of an "ideal" rule, see the seminal study by Kopits and Symansky (1998). Wren-Lewis (2011) presents a more recent overview of the literature on the deficit bias of governments and politicians.

³ See, e.g. Section III of IMF (2009) for the overview of the empirical evidence on the impact of fiscal rules.

investigating if actual compliance with rules alter their effect on volatility of fiscal policy. In the sense of the latter contribution to the literature, our paper is most closely related to the study by Reuter (2015) that looks at the effect of compliance with numerical fiscal rules and finds evidence that even if rules are not always adhered to they still help to steer fiscal policy variables towards their numerical limits.

We employ panel data for the EU28 countries that share a similar institutional fiscal environment shaped by the EU common fiscal governance framework over the period 1996–2015. In our econometric specification we regress the standard deviation of the discretionary fiscal policy shock (measure of fiscal policy stability) on alternative measures of fiscal rules. We account for heterogeneity in rule design by differentiating between budget balance rules that are defined in cyclically adjusted and nominal terms as well as expenditure rules setting ceilings that are independent of the current position in the business cycle. In order to investigate the effect of compliance with the rules, we include an interaction of the fiscal rule variable with a variable that accounts for compliance with it. We check the robustness of our estimates with respect to expanding the sample by including data of non-EU OECD countries as well as employing alternative time windows and alternative indicators of fiscal rules that account for their stringency. Finally, after establishing the relationship between the rules and fiscal policy volatility we investigate whether fiscal rules contribute to macroeconomic stability (directly or indirectly by lowering fiscal volatility).

The main findings of our study can be summarised as follows. We confirm that fiscal rules are effective means to limit fiscal policy volatility. In particular, the effect can be observed for "a-cyclical" rules, i.e. budget balance rules that set limits in cyclically adjusted terms and expenditure rules that restrict expenditure growth relative to potential GDP. Furthermore, the stabilising effect is stronger for more stringent rules. However, actual compliance with rules is not found to be significant. This might suggest that appropriately designed, stringent rules act as a benchmark for fiscal policy makers and the public. Thus, the rules seem to contribute to a more anchored and predictable fiscal policy even if they are not complied with year-to-year. Furthermore, we document that less volatile discretionary fiscal policy indeed improves macroeconomic stability. To this end, our conclusions support the view that rules, even if they are not adhered to year-to-year, can be effective in lowering fiscal volatility and hence contribute to macroeconomic stability.

The remainder of the paper is structured as follows. Section 2 summarises evidence from three streams of the related literature. Section 3 describes the methodology and the data employed in this study as well as the construction of the measure of fiscal policy shock. Section 4 gives an overview of national fiscal rules in the EU28 countries, examines their properties as well as assesses the compliance with the respective rules. Section 5 provides the estimation results using different measures of fiscal volatility. It also uncovers the effect of various rule characteristics as well as actual compliance with fiscal rules and checks the robustness of the baseline estimates. Finally, Section 6 concludes.

2. EVIDENCE FROM THE EMPIRICAL LITERATURE

From a theoretical point of view macroeconomic policy effects of fiscal rules are not straightforward. On the one hand, restrictions should limit volatility in the conduct of discretionary fiscal policy and thereby output volatility. On the other hand, rules might also limit the capacity to use discretionary fiscal policy for smoothing out business cycle fluctuations. The existing empirical evidence seems to lean in favour of the former effect of fiscal rules.

Using a large cross section of countries, Fatas and Mihov (2003) argue that the presence of political constraints accounts for a large portion of the cross-country variation in the use of fiscal policy. Furthermore, they show that the aggressive use of discretionary fiscal policy amplifies business cycle fluctuations. Similarly, Fatas and Mihov (2006) find that fiscal policy restrictions are effective means of reducing policy volatility in the US states that in turn contributes to macroeconomic stability. They show that the effect of rules on output stability is intermediated by less aggressive use of discretionary fiscal policy. The two more recent papers by Sacchi and Salotti (2015) and Badinger and Reuter (2017) investigate the effect of fiscal rules focusing on two alternative channels. Sacchi and Salotti (2015) argue that fiscal rules may render fiscal volatility less destabilising if fiscal shocks are procyclical, so that the presence of fiscal rules reduces the destabilising procyclicality in the conduct of discretionary fiscal policy. They find empirical evidence for this effect and show that budget balance rules are more efficient than other types of rules in reducing procyclicality. Badinger and Reuter (2017) argue that there is an indirect negative relation between fiscal rules and output volatility via the rules' stabilising effect on fiscal policy volatility (similarly as in Fatas and Mihov (2003; 2006)). Furthermore, more stringent fiscal rules that have strong legislative support, broader coverage and formal enforcement procedure in place appear to be more efficient in reducing fiscal policy volatility than less rigorous rules.

While the literature on the general effects of the stipulation of fiscal rules is quite large, there are only few studies looking at the design features of fiscal rules and even fewer looking at actual compliance with them. One of the first comprehensive studies that investigates compliance with fiscal rules is Reuter (2015). The paper looks at 23 fiscal rules, enacted in 11 EU countries from 1994 to 2012. It shows that even if rules are not complied with they help to steer fiscal policy variables towards their numerical limits. Fiscal rules seem to act as a benchmark for both fiscal authorities and the general public. Cordes et al. (2015) show that countries comply with expenditure rules more often than with budget balance rules, particularly if they cover a part of public finances that is under direct government control and if the rules are enshrined into legislation.

Reuter (2017) looks at the causes of non-compliance with national fiscal rules in 20 EU countries from 1995 to 2015. The results indicate that macroeconomic factors are irrelevant in explaining adherence to fiscal rules, while some political factors, like government fragmentation or decentralisation, turn out to be significant. In a country- specific environment, Delgado-Téllez et al. (2017) analyse determinants of non-compliance with Spanish regional fiscal deficit targets over the period 2002–2015. They show that the strength of fiscal rules in Spanish regions does not have an impact on the frequency of non-compliance or at least it is not robust across different estimation procedures.

3. ESTIMATION FRAMEWORK

3.1. Fiscal rules and fiscal policy volatility

Following Fatas and Mihov (2006) and Badinger and Reuter (2017), we estimate the effect of stipulating fiscal rules on the volatility of discretionary fiscal policy:

$$\sigma_{i,t}^{\widehat{e^{discr.fp}}} = Z'_{i,t} \,\partial + \alpha_1 F R_{i,t} + \mu_i + \omega_{i,t} \tag{1}$$

where $\sigma_{i,t}^{\varepsilon^{discr.fp}}$ is fiscal policy volatility measured as the standard deviation of discretionary shocks in one of four different fiscal aggregates: primary expenditure, public consumption, the sum of public consumption and public investment and primary balance. The calculation of the discretionary shocks is described in detail later in this section. $Z'_{i,t}$ is a vector of control variables with the respective coefficient ∂ . μ_i denotes country-fixed effects and $\omega_{i,t}$ - the idiosyncratic error term. $FR_{i,t}$ is a measure of fiscal rules.

Estimations are based on data of the EU28 countries, covering the period 1996–2015. The sample is split into four non-overlapping 5-year periods: 1996-2000, 2001-2005, 2006-2010 and 2011-2015 in total comprising 110 observations. As a robustness check we test the sensitivity of our results with respect to the length of these time windows. Our sample is restricted to 1996-2015 as data availability for earlier years is low, particularly for the Member States that joined the EU in 2004 or later. Furthermore, also the number of fiscal rules that were in force in the early 1990s is small. In our robustness checks we also augment our dataset with non-EU OECD countries⁴ which gives us 45 additional observations at the cost of using a more heterogeneous country sample.

We consider two different types of fiscal rules: budget balance rules that set restrictions on the level of fiscal balances either in nominal or cyclically adjusted terms; and expenditure rules aimed at keeping the expenditure-to-GDP ratio below a certain threshold or restricting changes in government expenditure. We estimate equation (1) for each type of rule separately.

As in Fatas and Mihov (2006) and Badinger and Reuter (2017), we employ various measures of fiscal rules. First, we include the share of years within each 5-year period in which a country had a fiscal rule in place.⁵ In addition, to account for the heterogeneity of fiscal rules as well as to ensure robustness of our estimates we use measures of the stringency of fiscal rules based on the POSET theory (calculated by Badinger and Reuter (2015)) as well as the IMF fiscal rules index (with the methodology laid out in Schaechter et al. (2012)).

Following our baseline estimations described above, we also explore whether the design of fiscal rules, specifically if they have "a-cyclical" properties, matters. Therefore, we analyse the sign and significance of the following interaction terms:

a) For budget balance rules: the share of years in which the rule is set in cyclically adjusted or structural terms. The rules based on basic nominal variables may have a destabilising effect on fiscal policies due to ill-timed binding constraints. However, cyclically adjusted variables should allow the full operation of automatic stabilizers and deviations of nominal balances from targeted levels at different stages of the business cycle.

b) For expenditure rules: the indicator of whether the limit set by the rule partials out the current state in the business cycle, e.g. by relating expenditures to developments in *potential* output opposed to nominal output or other measures. Expenditure rules that base their limit, e.g. only on nominal output (growth), may turn policy destabilising. When the economy is booming, such a rule would provide extra room for additional spending and the contrary would be the case during recessions.

Setting budget balance ceilings in cyclically adjusted terms and/or tying budget expenditure to potential output could prevent procyclical behaviour and might therefore contribute to policy stabilisation.

Next, we include interactions with a compliance variable to explore the impact of compliance with fiscal rules on their effect on fiscal volatility. Within each 5-year period the variable is equal to the number of years in which the rule was adhered to over the

⁴ These countries are: Australia, Canada, Chile, Iceland, Israel, Japan, Mexico, New Zealand, Norway, Korea, Switzerland, Turkey and the US. ⁵ For example, if a country has a fiscal rule in force in two out of five years, the variable takes the value of 0.4.

number of years the rule was in force.⁶ For budget balance rules we also investigate whether the distance between the realised value and the numerical constraint has any influence on the fiscal policy volatility.

There might be concerns that fiscal rules are endogenous to fiscal conditions if governments introduce them inspired by fiscal performance. In a cross-section setting reverse causality could be addressed by instrumenting fiscal rules using institutional or political variables, such as the nature of the political system and monetary policy setup. However, in a panel framework this gets more complicated due to the lack of time variation in such instruments, particularly in a panel as ours where the time dimension is represented by time windows. Still, we tried employing variables (government fragmentation, checks and balances and inflation targeting) identified as fiscal rules determinants by Badinger and Reuter (2017) in their broad panel of 71 countries. However, in our study they turn out to be uncorrelated with fiscal rules as we use a much more homogeneous group of countries and a more limited time span further divided into 5-year time windows.

Importantly, Badinger and Reuter (2017) show that the Hausman test does not reject the null hypothesis of exogeneity of fiscal rules when these are used as determinants of fiscal volatility. They argue that compared with fiscal balances, volatility of discretionary fiscal policy is a much less reported and debated indicator, making it unlikely that governments introduce or strengthen fiscal rules in response to changes in fiscal policy volatility. Similarly, in a comprehensive study of fiscal rules and their effect on fiscal balances, Debrun et al. (2008) reveal that the risk of reverse causality between fiscal rules and fiscal discipline is relatively small. They estimate fiscal reaction functions using least squares and instrumental variable approaches and find almost identical magnitudes of the impact of fiscal rules. Therefore, we proceed without instrumenting fiscal rules while acknowledging that endogeneity may still be relevant in the context of our study and represents an avenue for future research.

Following Fatas and Mihov (2006), we include the following control variables: GDP per capita, population and dependency ratio. In addition, we capture institutional characteristics by country fixed effects and use a government fragmentation index, calculated as the sum of the squared seat shares of the parties in government.⁷

3.2. Fiscal rules and output volatility

Finally, we investigate whether fiscal rules contribute to macroeconomic stability in a direct or indirect way. To this end, we estimate the following regression:

$$\ln(\sigma_{i,t}^{\Delta GDP \ pc}) = \beta_1 \sigma_{i,t}^{\widehat{\mathcal{E}^{discr.fp}}} + \beta_2 FR_{i,t} + W'_{i,t} \ \theta + \eta_i + \vartheta_{i,t}$$
(2)

where vector $W_{i,t}$ is comprised of the control variables, which have been identified as determinants of output volatility in the previous literature: government size⁸, calculated as primary expenditure ratio to GDP (emphasised by Galí (1994), Fatas and Mihov (2001) and Debrun et al. (2008) among others), log of real GDP per capita⁹ (see, e.g. Acemoglu and

⁶ If a rule was in force in all five years and was adhered to in three years out of the five, the compliance variable takes the value of 0.6.

⁷ Most of the institutional variables suggested as determinants of fiscal volatility by Fatas and Mihov (2003) and Badinger (2009), such as a measure of political regime, the number of elections or degree of fiscal decentralisation, become useless in a panel setting due to low or lack of time variation.

⁸ Larger governments exhibit smaller output fluctuations than smaller governments because they have more possibilities and larger automatic stabilizers to smooth out business cycles.

⁹ Richer countries display a more diversified sectoral decomposition of output that reduces its variability.

Zilibotti (1997) and Greenwood and Jovanovic (1990)) and trade openness¹⁰, expressed by the sum of exports and imports relative to GDP (Rodrik (1998), Easterly et al. (2001) as well as di Giovanni and Levchenko (2009)). We use average values of these variables over the corresponding time windows. As explained in detail in Rodrik (1998), government size might be affected by our dependent variable – output volatility. Thus, government size needs to be treated as endogenous. Following the existing literature (e.g. Badinger (2009), Sacchi and Salotti (2015)), 5-year average values of the following variables are used as instruments for government size: log of the total population, dependency ratio and urbanisation rate.

First, in order to assess whether there is a direct relation between fiscal restrictions and output volatility, we estimate equation (2) by least squares and including *FR* directly in the regression. Then, to assess the presence of only an indirect effect, we estimate equation (2) using a 2SLS procedure, where we exclude *FR* from the main equation and use it instead as an instrument for fiscal volatility among the other above-mentioned determinants.¹¹

Table A1 in the Appendix gives an overview of the variables used in our analysis and their sources. Table A2 presents the corresponding descriptive statistics.

3.3. Estimation of fiscal policy volatility

We do not deviate from the existing literature in calculating the volatility of discretionary fiscal policy and focus on purely *exogenous* fiscal policy shocks, i.e. looking at the part of the change in real fiscal variables that does not emanate from cyclical fluctuations either automatically or endogenously.¹² To this end, we cannot rely on cyclically adjusted government series (Badinger (2009)). We base our calculations on the standard methodology employed, e.g. in the papers by Sacchi and Salotti (2015), Badinger (2009), Fatas and Mihov (2003; 2006), Agnello at al. (2013) and Afonso et al. (2010).¹³ We estimate equation (3) to obtain quantitative estimates of the fiscal shocks as the residual ($\varepsilon_{i,t}$), i.e. the part of the change in a government series that is not attributable to a country's position in the business cycle or changes in other control variables:

 $\Delta \log(G_{i,t}) = \alpha_i + \beta_1 \Delta \log(G_{i,t-1}) + \beta_2 \Delta \log(Y_{i,t}) + \beta_3 \pi + \beta_4 \pi^2 + \beta_5 trend + \varepsilon_{i,t}$ (3)

where $G_{i,t}$ is the annual growth rate of a fiscal variable in real terms for country *i* in period *t*, *Y* is real GDP, π is the rate of inflation included alongside its squared value and *trend* is a linear time trend. Inflation is included to capture the possibility that in some countries several expenditure items are not inflation-adjusted and might be affected by inflation shocks at least in the short-run. We consider four alternative fiscal policy variables to measure the discretionary fiscal policy shock: primary expenditure, public consumption, sum of public consumption and investment and primary balance.¹⁴ To address endogeneity

¹⁰ Output volatility tends to be larger in countries that are more open to international trade and hence exposed to fluctuations in the global market and external shocks.

¹¹ Note that the urbanisation rate and trade openness are also included as controls for fiscal volatility in equation (1) to ensure that the variables used as fiscal volatility determinants in equation (1) and as instruments for fiscal volatility in first-stage regressions of equation (2) are identical. The inclusion of these two variables does not have any significant impact on the estimation results of equation (1).

¹² In the latter case, the government is deliberately applying pro- or counter-cyclical fiscal policies in response to the business cycle. For a detailed discussion of fiscal balance decomposition, see Galí et al. (2003).

¹³ One of the first methods to construct measures of discretionary fiscal policy was laid out in Blanchard (1993), also applied by Alesina et al. (1995). Fatas and Mihov (2003) show that the employment of this approach is equivalent to what they do in their paper and what we follow in our study.

¹⁴ All of the expenditure aggregates are used in real terms, whereas budget balance variables are expressed in % of GDP.

of real GDP, we instrument real GDP growth by its two lags, an oil price index and lagged inflation.

We estimate equation (3) for each of the four alternative fiscal variables using both the 2SLS and GMM approaches¹⁵ in the panel of EU28 countries.¹⁶ Since in the presence of a lagged dependent variable 2SLS may result in inconsistent estimates, we employ GMM estimated fiscal shocks in the baseline. However, to check the robustness of our results with respect to the choice of the econometric approach, we also present the results based on 2SLS estimates instead of GMM.

We obtain the residuals from equation (3) and calculate the standard deviation over our four non-overlapping 5-year periods for each country. This serves as our indicator of fiscal volatility in both equation (1) and equation (2). Tables A3 and A4 in the Appendix show that the positive correlation between different measures of fiscal volatility is notable, in particular when variables are derived from regressions estimated by GMM.

4. DATA ON FISCAL RULES

4.1. Presence and characteristics of fiscal rules

In this study we use information on fiscal rules from two major datasets compiled by the EC (2016) and the IMF (2017).¹⁷ The IMF dataset covers national and supranational fiscal rules in 96 countries from 1985 to 2015, while the EC database considers only the national fiscal rules that have been in force since 1990 across EU Member States. Both datasets provide definitions and descriptions of fiscal rules as well as information on their properties (e.g. their legal basis, coverage, monitoring body, escape clauses and enforcement procedure).

The sample of this study includes national fiscal rules¹⁸ in the EU28 countries from 1996 to 2015. Due to data availability we focus only on the rules that are set at central or general government levels and have already come into force or were in force but are not anymore (rather than those that are merely planned in the future). We consider two types of numerical fiscal rules: budget balance rules and expenditure rules. Table 1 presents an overview of how the rules included in our study are distributed across countries and rule characteristics.

			Budget balance rule (%)	Expenditure rules (%)
Rule in place	% of countries	(1)	89	64
	% of observations	(2)	34	34
Statutory base	Constitution	(3)	16	2
	Legal act	(4)	45	34

Table 1Main characteristics of fiscal rules

¹⁵ We use a GMM-difference estimator (Arellano and Bond (1991)) not only because the results of the Sargan test based on GMM-system (Arellano and Bover (1995), Blundell and Bond (1998)) estimations are not satisfactory, but also because there is only low persistency in the change of fiscal variables.

¹⁶ Employing country-by-country estimation would entail few degrees of freedom due to the very short time series (we deal with a maximum of 20 observations per country). Therefore, in this paper we present results based on the panel estimates of fiscal shocks, even though those might be heterogeneous across EU Member States. However, Sacchi and Salotti (2015) show that the residuals of this model estimated country by country are highly correlated with the residuals obtained with panel estimates.

¹⁷ A detailed description of the earlier version of the IMF dataset is provided in Schaechter et al. (2012).

¹⁸ We do not consider supranational fiscal rules since i) in our sample, they have been introduced with exactly the same properties simultaneously in many countries and thus cannot explain cross-country differences, ii) they have to a large extent been incorporated into national legislation.

	Coalition agreement	(5)	32	42
	Political commitment	(6)	8	22
Monitoring outside	Exists	(7)	37	35
government	Does not exist	(8)	63	65
Enforcement procedure	Exists	(9)	36	28
	Does not exist	(10)	64	72
Escape clause	Exists	(11)	28	6
	Does not exist	(12)	72	94
Budget balance rule definition	In cyclically adjusted or structural terms	(13)	49	_
	Other	(14)	51	_
Expenditure rule	Potential growth benchmark	(15)	_	8
definition	Other	(16)	_	92

Sources: EC (2016) and IMF (2017).

Notes. The first row of the table presents the share of countries (%) that have had a fiscal rule in at least one year from 1996 to 2015. The rest of the figures included in the table reflect the number of observations (% of total) with the respective characteristic.

Overall, national fiscal rules seem to be part of fiscal frameworks in most of EU Member States. Nearly all countries in the sample have introduced a budget balance rule in one way or another, whereas expenditure rules are relatively less common (89% and 64% of countries respectively). More than half of the budget balance rules and approximately a third of expenditure rules are enshrined either in legal acts or constitutions. Generally, expenditure rules have a somewhat weaker legal status as a majority of them are set out in coalition agreements or political commitments. External monitoring bodies exist in approximately a third of the observations. A similar share has a formal enforcement procedure. Escape clauses are embedded in approximately a quarter of budget balance rules, a feature less common for expenditure rules.

In nearly half of the observations we consider in our study budget balance rules are set in cyclically adjusted (or structural) terms, i.e. a limit is formulated in a way to allow the full operation of automatic stabilizers. Only three countries had a cyclically adjusted budget balance rule from the very beginning of our sample. These countries are Denmark (in force since 1992), Finland (1999) and Sweden (2000). Finland and Sweden introduced their numerical fiscal rules as part of a comprehensive reform of their fiscal frameworks in response to a severe economic and fiscal crisis they experienced in the early 1990s. In the rest of the EU, the introduction of a cyclically adjusted rule took place after the crisis of 2007–2008. The adoption of numerical national fiscal rules, including the cyclically adjusted ones, was facilitated by the approval of the EU requirements for budgetary frameworks of the Member States¹⁹ in 2011. As a result, the number of countries that set their budget balance limit in cyclically adjusted terms grew to 22 by 2015.

Similarly, expenditure rules that tie government outlays to the developments in potential GDP (rather than setting a threshold for expenditure growth or its ratio in relation, e.g. to nominal GDP) have also come into fashion only recently. In particular, five EU countries introduced such rules since 2006, i.e. Bulgaria, Denmark, Croatia, Poland and Latvia. Among these countries only Latvia introduced such an expenditure rule from scratch, whereas Bulgaria and Denmark added it to the already existing nominal rule.²⁰ Croatia and

¹⁹ Council Directive 2011/85/EU of 8 November 2011 on requirements for budgetary frameworks of the Member States.

²⁰ These nominal rules imposed a ceiling on the expenditure-to-GDP ratio (Bulgaria) or the level of expenditure (Denmark).

Poland²¹ replaced the nominal expenditure rule with a rule respecting the business cycle. Within such rules, additional spending (i.e. beyond that determined by potential GDP developments) is allowed only when it is financed by specific discretionary revenue measures.

4.2. Compliance with fiscal rules

Finally, we look at compliance with national fiscal rules. Staying below the numerical ceilings has often been named as one of the key elements when defining a successful fiscal framework. Nevertheless, previous evidence shows that compliance *per se* might not be necessary to reap positive effects of fiscal rules (Reuter (2015)). To further explore the economic implications of fiscal rules, we look at a subsample of fiscal rules for which a numerical ceiling and compliance with it can be measured. It has to be noted that we only observe statistical compliance with fiscal rules as opposed to *de jure* legal compliance, which would need some kind of judgment by an external monitoring body, e.g. by a fiscal council. We draw on Reuter (2015) database which consists of numerical limits (calculated on the basis of the respective legal documents) and actual outcomes of the constrained budgetary positions for the EU28 countries from 1996 to 2013. Using the same methodology, we expand the dataset to include the most recent years. Overall, it seems that breaching national fiscal rules is not an extraordinary event. We observe that countries have met the limits imposed by fiscal rules in less than half of the years. The incidence of (non-)compliance is broadly similar for both types of rules (41% for budget balance rules and 43% for expenditure rules). Judging by the simple comparison of the mean POSET values, rules that were complied with do not appear notably different in terms of stringency from the rules that were not complied with. It appears that compliance with budget balance rules that partial out cyclical developments is somewhat higher (46%) than compliance with other types of budget balance rules (39%). As for expenditure rules that set expenditure ceilings based on potential output developments, the number of observations for this type of rules is too small to draw meaningful conclusions.

Table 2 reports mean values of fiscal volatility for various degrees of compliance with fiscal rules. Discretionary fiscal policy indeed appears more volatile when there is no national fiscal rule in place. Hence, setting a budget balance or expenditure ceiling seems to promote more stable discretionary fiscal policy. However, the design of the rules matters, especially in connection with compliance. For rules which are designed in an "a-cyclical" way, e.g. cyclically adjusted budget balance rules, higher compliance is associated with less volatile fiscal policy, while for other rules the opposite is the case and lower fiscal volatility is observed with less compliance. This supports the intuition that if a country complies with a rule with pro-cyclical design, volatility increases. In Subsection 5.3 we test the relationship between rule design, compliance and volatility more formally in a regression framework.

	Primary expenditure	Public consumption	Public consumption and investment	Primary balance
No rule	0.787	0.821	0.766	0.756
Budget balance rule	0.529	0.424	0.374	0.523
Complied with <50%	0.530	0.413	0.368	0.504
Complied with >50%	0.528	0.451	0.388	0.571

Table 2Mean fiscal volatility for different levels of compliance with fiscal rules

²¹ In Poland, a permanent expenditure rule, limiting the growth of public expenditure to trend GDP growth, took effect in 2015. However, later it was modified to allow more flexibility to increase spending (IMF (2017)).

CAB rule	0.401	0.323	0.293	0.439
Complied with <50%	0.406	0.331	0.301	0.430
Complied with >50%	0.391	0.309	0.279	0.455
Other budget balance rules	0.728	0.580	0.500	0.655
Complied with <50%	0.685	0.514	0.452	0.596
Complied with >50%	0.903	0.841	0.689	0.890
Expenditure rules	0.328	0.282	0.250	0.360
Complied with <50%	0.237	0.224	0.192	0.293
Complied with >50%	0.516	0.401	0.368	0.500

Sources: data on compliance from Reuter (2015) for the period 1995–2013 and authors' calculations for 2014–2015. In order to assess the compliance, we calculate numerical limits based on the description of fiscal rules in IMF's (2017) dataset and respective legal documents.

Note. Mean values of fiscal volatility were calculated by using shares of years (within the 5-year periods) when a fiscal rule was in force as weights.

5. ESTIMATION RESULTS

In this section, we present the main findings of this study. First, we report the baseline results on the impact of fiscal rules on fiscal volatility (see Subsection 5.1). Second, we investigate the influence of stabilisation properties of fiscal rules (see Subsection 5.2). Next, we analyse if actual compliance with fiscal rules changes the effects (see Subsection 5.3). Afterwards, we present some robustness checks of our results (see Subsection 5.4). Finally, we examine whether fiscal rules have an effect on macroeconomic stability (see Subsection 5.5).

5.1. Baseline results

In this study, we aim at exploring whether and how fiscal policy stability is affected by the existence of national numerical fiscal rules.

Table 3 documents the baseline results that show that lower volatility of discretionary fiscal policy²² is associated with the presence of budget balance rules. This finding appears to be robust across various fiscal aggregates, albeit with somewhat different magnitudes of estimated coefficients. In turn, expenditure rules do not seem to exert a significant effect on fiscal volatility.

For the interpretation of the magnitude of the effect of fiscal rules on fiscal volatility, we first, as an illustrative example, look at the results for primary expenditure and budget balance rules. An increase in the budget balance rule variable by one standard deviation (which corresponds to having a rule for two additional years) results in a 0.2 percentage point lower fiscal volatility²³ (which has a mean of 0.7 as presented in Table A2 in the Appendix).

Extending this example for other fiscal aggregates, we see that the calculated effect of having a budget balance rule on fiscal volatility is relatively stable across different specifications. An increase in the presence of budget balance rule by 1 standard deviation on average decreases fiscal volatility by around 0.2–0.3 percentage point.

 $^{^{22}}$ In the baseline regression we employ fiscal volatility measures estimated using a GMM approach, while the results based on estimations using fiscal volatility estimated with 2SLS are very similar and are provided in Tables A6–A8 in the Appendix.

 $^{^{23}}$ The value 0.221 is obtained by multiplying the coefficient of budget balance rule variable (-0.517) and the standard deviation in this variable (0.428; see Table A2 in the Appendix).

	Primary e	xpenditure	Public cor	nsumption	Public cor and inv	nsumption estment	Primary	balance
Dep. var.: Fiscal volatility								
Budget balance rule	-0.517*		-0.647**		-0.560*		-0.471*	
	(0.269)		(0.290)		(0.285)		(0.264)	
Expenditure rules		-0.082		-0.180		-0.199		0.061
		(0.220)		(0.240)		(0.217)		(0.265)
Population size	11.261***	10.271***	10.336***	8.963***	10.629***	9.369***	9.020***	8.352***
	(3.060)	(3.002)	(3.007)	(2.852)	(2.896)	(2.689)	(2.824)	(2.700)
Dependency ratio	-0.025	-0.030	-0.013	-0.018	-0.010	-0.013	0.016	0.006
	(0.028)	(0.029)	(0.027)	(0.028)	(0.025)	(0.025)	(0.032)	(0.030)
Urbanisation rate	0.002	-0.007	0.029	0.020	0.009	0.001	0.052	0.041
	(0.029)	(0.030)	(0.026)	(0.028)	(0.025)	(0.028)	(0.035)	(0.033)
Government	-1.301	-1.569	0.703	0.387	0.315	0.053	-0.686	-0.964
fragmentation	(1.308)	(1.332)	(1.480)	(1.586)	(1.401)	(1.503)	(1.140)	(1.092)
GDP per capita	1.993***	1.763***	1.485**	1.221*	1.170*	0.955	1.271*	1.020
	(0.569)	(0.582)	(0.710)	(0.719)	(0.662)	(0.679)	(0.678)	(0.681)
Trade openness	-0.006	-0.006	-0.005	-0.006	-0.005	-0.005	-0.005	-0.005
	(0.004)	(0.004)	(0.004)	(0.005)	(0.004)	(0.004)	(0.004)	(0.004)
R-sq within	0.271	0.230	0.200	0.147	0.203	0.152	0.209	0.174

Table 3Effects of fiscal rules on fiscal volatility

Notes. * indicates significance at a 10% level, ** indicates significance at a 5% level, *** indicates significance at a 1% level. The sample covers the EU28 countries and four 5-year time windows over the period 1996–2015 (110 observations). Cross-sectional fixed effects are also accounted for. Coefficient t-test results are based on heteroskedasticity robust standard errors. These standard errors are presented in parentheses.

5.2. Stabilising properties of fiscal rules

As fiscal rules are sometimes claimed to be procyclical, many countries introduced rules which partial out the current position in the business cycle and are thus in a way "a-cyclical". We study two of the potential stabilising features of fiscal rules. First, we make a distinction between budget balance rules, which constrain budget balances in nominal terms and those which constrain balances in cyclically adjusted or structural terms. In the regression, this is accomplished by interacting the budget balance rule with a variable that represents the share of years a rule is set in cyclically adjusted or structural terms. Second, similarly, we make a distinction between the expenditure rules that tie expenditure growth to potential output developments and other expenditure rules, which in most cases impose a threshold on the expenditure-to-GDP ratio or establish expenditure ceilings which depend on current revenue or growth.

It turns out that the baseline effects of budget balance rules on fiscal volatility, found in Subsection 5.1, are driven by the rules which limit CABs rather than nominal ones. This result is robust across all definitions of fiscal volatility (see Table 4) and the magnitude of the effect of such structural balance rules is also somewhat stronger, with an overall decrease in fiscal volatility of 0.3–0.4 percentage point. Thus, the results found in the literature so far might need to be interpreted with caution as not all budget balance rules seem to decrease fiscal volatility, but rather only the ones that are defined in cyclically adjusted or structural terms. Nevertheless, we also do not find the opposite result for budget balance rules constraining nominal measures. They only turn out to be insignificant while they are not found to contribute to higher volatility.

With respect to expenditure rules, contrary to the baseline results, we find significant effects if we take into account how the ceiling of such rules is set. The coefficients for the interaction term, indicating if the ceiling of an expenditure rule is tied to potential output, appear statistically significant for almost all measures of fiscal volatility. This, together with the results of the F-test and the joint significance test, implies that expenditure rules in general seem to be neutral with respect to fiscal policy volatility, but reduce volatility if they base their numerical limit on a measure related to potential output (as opposed to current or nominal output). The magnitude of the effect of such expenditure rules is similar to the effect of structural balance rules, i.e. between 0.2 and 0.4 percentage point. However, this result should be treated with a considerable portion of caution as the number of this type of rules is still relatively small.

	Primary expenditure		Public consumption		Public con and invo	sumption estment	Primary balance	
Dep. var.: Fiscal volatility								
Budget balance rule	-0.307		-0.269		-0.282		-0.194	
	(0.313)		(0.296)		(0.312)		(0.284)	
Budget balance rule \times	-0.415*		-0.749***		-0.551**		-0.550**	
CAB rule	(0.228)		(0.271)		(0.249)		(0.261)	
Expenditure rules		-0.050		-0.102		-0.141		0.119
		(0.213)		(0.216)		(0.198)		(0.261)
Expenditure rules \times		-0.345		-0.836**		-0.616*		-0.614**
Potential output benchmark		(0.308)		(0.371)		(0.332)		(0.275)
F-test	13.30***	4.93**	17.99***	6.44**	15.99***	6.27**	7.86***	3.29*
Joint-sign test	6.66***	3.07*	9.16***	3.37**	8.17***	3.23*	4.01**	5.51***
R-sq within	0.289	0.237	0.255	0.183	0.239	0.180	0.240	0.198

Table 4Effect of fiscal rules on fiscal volatility (by type of rule)

Notes. * indicates significance at a 10% level, ** indicates significance at a 5% level, *** indicates significance at a 1% level. The sample covers the EU28 countries and four 5-year time windows over the period 1996–2015 (110 observations). Controls as well as cross-sectional fixed effects are also included but not reported for the sake of brevity. Coefficient t-test results are based on heteroskedasticity robust standard errors. These standard errors are presented in parentheses. The null of F-test means that the sum of coefficients on a fiscal rule variable and on its interaction is equal to zero. The null of joint significance test means that both a coefficient on a fiscal rule variable and on its interaction are equal to zero.

5.3. Compliance with fiscal rules

In the past years, strong emphasis has been put on strengthening fiscal frameworks to increase compliance with fiscal rules. While in general higher compliance rates might increase the effects of fiscal rules on debt sustainability and fiscal policy stabilisation, rules might also have an effect without being adhered to year-to-year. Furthermore, the effect on fiscal volatility might differ, depending on a specific rule. For example, a structural budget balance rule might be counter-cyclical if always complied with but turn neutral or procyclical if it is sometimes not complied with. For nominal budget balance rules the opposite might be true, with them possibly being pro-cyclical if always complied with.

The estimation results reported in Table 5 show no evidence that compliance with rules affects the impact of fiscal rules on fiscal volatility. The compliance variable is always insignificant, while the coefficients of the rule variables as well as the test statistics remain broadly in line with the previous results. These results might suggest that there is an effect of fiscal rules on fiscal volatility which is not due to the strict constraint for fiscal policy, but rather because they work as a benchmark or anchor for fiscal policy which reduces volatility even if the rules are not always complied with. Nevertheless, we cannot rule out

that rules might have even stronger effects if they are always complied with (which we do not observe in our sample).

	Prima	ry exper	nditure	Public	consum	ption	Public and	consun	nption nent	Prin	Primary balance	
Dep. var.: Fiscal volatility												
Budget balance rule	-0.459	-0.305		-0.633**	-0.253		-0.548*	-0.269		-0.478*	-0.151	
	(0.285)	(0.317)		(0.297)	(0.300)		(0.291)	(0.316)		(0.274)	(0.285)	
Expenditure rules			-0.092			-0.142			-0.143			0.088
			(0.197)			(0.223)			(0.186)			(0.287)
Budget balance rule ×		-0.421			-0.794**	•		-0.586**	k		-0.662**	k
CAB rule		(0.256)			(0.313)			(0.286)			(0.288)	
Budget balance rule \times	-0.318			-0.081			-0.065			0.036		
Compliance	(0.383)			(0.356)			(0.286)			(0.426)		
Expenditure rules ×			0.040			-0.159			-0.230			-0.110
Compliance			(0.402)			(0.454)			(0.427)			(0.436)
Budget balance rule \times		0.019			0.153			0.122			0.385	
CAB rule ×		(0.298)			(0.362)			(0.278)			(0.419)	
Compliance												
F-test	6.56**	2.05	0.01	4.76**	3.01*	0.38	4.54**	2.39	0.72	1.72	0.70	0.00
Joint-sign test	5.31**	2.08	0.06	3.75**	3.97**	0.27	3.79**	3.05**	0.45	2.08	2.33*	0.05
R-sq within	0.279	0.290	0.230	0.201	0.257	0.142	0.203	0.240	0.155	0.210	0.247	0.175

Table 5Effect of compliance with fiscal rules on fiscal volatility

Notes. * indicates significance at a 10% level, ** indicates significance at a 5% level, *** indicates significance at a 1% level. The sample covers the EU28 countries and four 5-year time windows over the period 1996–2015 (110 observations). Controls as well as cross-sectional fixed effects are also included but not reported for the sake of brevity. Coefficient t-test results are based on heteroskedasticity robust standard errors. These standard errors are presented in parentheses. The null of F-test means that the sum of coefficients on a fiscal rule variable and on its interaction is equal to zero. The null of joint significance test means that both a coefficient on a fiscal rule variable and on its interaction are equal to zero.

5.4. Robustness checks

In the following section, we present extensive robustness checks of our estimation results discussed above. The summary of the robustness analysis for the case when fiscal volatility is estimated based on public consumption series²⁴ is presented in Table 6.²⁵

First, we check the sensitivity of our results to the country composition of our sample, by augmenting it with the OECD countries that are not part of the EU. Their inclusion allows us to increase the number of observations up to 155. The main results, including the dampening effect of fiscal rules on fiscal volatility, are confirmed also in this larger country sample. However, the magnitude of the effects of fiscal rules on fiscal volatility becomes slightly smaller. This is not surprising as the heterogeneity of the country sample, e.g. in institutional and cultural aspects is increased when moving away from EU Member States.

Second, we split our sample into smaller, 4-year, periods in order to check if our baseline results are influenced by the number of years over which the included variables are averaged and standard deviations are calculated. Again, the main results are broadly confirmed.

Third, we test alternative measures of fiscal rules: i) a dummy variable that takes the value one if there is a fiscal rule in place in at least one year within each 5-year period and zero otherwise; ii) fiscal rule indices (POSET index, Badinger and Reuter (2015) and an index constructed by the IMF in Schaechter et al. (2012)) to also account for the stringency and coverage of fiscal rules. Independently of how the presence of fiscal rules is measured the estimated negative effect of fiscal rules remains statistically significant and the main conclusions are confirmed.²⁶ The estimation results using indices of the stringency of fiscal rules indicate that more stringent fiscal rules bring about a larger reduction of fiscal volatility. The interaction terms that capture the stabilising properties of budget balance and expenditure rules remain significant, although the level of significance is somewhat lower. The effects of (non-) compliance remain insignificant. The latter finding is reinforced and further strengthened when we take into account how far the actual fiscal outturn is from the numerical constraint imposed by the rule (see Table 6, row 4). Thus, it seems that regardless of whether the ceiling is missed with a narrow or a wide margin, budget balance rules can still serve as an anchor for fiscal policy makers.²⁷

Finally, we look at the robustness of our results with respect to the econometric approach. Tables A5–A7 in the Appendix reproduce the estimations presented in the main part of this study using fiscal volatility estimated with 2SLS instead of GMM. Table A8 displays the estimation results for fiscal volatility when both budget balance and expenditure rules are included in a single regression simultaneously in order to avoid the possibility that by having both budget balance and expenditure rules at place the effect of a rule excluded from the regression is caught by a rule included in the regression. Overall, the results in both econometric exercises remain very similar and qualitatively robust with respect to the baseline.

²⁴ The public consumption series is chosen to illustrate the results of the robustness analysis due to the comparability of this series across all OECD countries (its construction is based on the same methodology). However, the results for other fiscal aggregates are available upon request.

²⁵ The rest of the estimated regressions are not shown for the sake of brevity. They are broadly in line with the robustness checks presented in Table 6 and are available upon request.

²⁶ The coefficient values and therefore the magnitudes of fiscal rule effects are not comparable between the baseline estimation and the estimation using POSET/IMF indices as those are measured in different units.
²⁷ We also differentiate between positive and negative distances to account for the possible asymmetric response if

²⁷ We also differentiate between positive and negative distances to account for the possible asymmetric response if the limit is missed or overachieved by the same margin; however our conclusions remain unchanged. These results are not reported but available upon request.

	Baseline	OECD	4-year	Rule	POSET	IMF
			periods	dummy		
1	-0.647***	-0.404**	-0.358**	-0.549***	-1.317***	-0.250***
2	-0.269	-0.153	-0.019	-0.300	-0.467	-0.100
	-0.749***	-0.493***	-0.639***	-0.396**	-1.112**	-0.189**
3	-0.633**	-0.362**	-0.346**	-0.586***	-1.301***	-0.249***
	-0.081	-0.152	-0.053	0.126	-0.085	-0.007
4	-0.648**	-0.405**	-0.379**	-0.554***	-1.315***	-0.173**
	-0.004	-0.004	-0.032	0.049	0.012	0.016
5	-0.180	-0.071	-0.012	-0.077	-0.608	-0.138
6	-0.102	0.018	0.135	0.017	-0.419	-0.097
	-0.836**	-0.688**	-0.756***	-0.782**	-1.387*	-0.221*
7	-0.142	х	0.106	0.005	-0.394	-0.088
	-0.159	Х	-0.372	-0.186	-0.588	-0.122
	1 2 3 4 5 6 7	Baseline 1 -0.647*** 2 -0.269 -0.749*** -0.633** 3 -0.633** -0.081 -0.081 4 -0.648** -0.004 -0.180 5 -0.180 6 -0.102 -0.836** -0.142 -0.159 -0.159	Baseline OECD 1 0.647*** 0.404** 2 0.269 0.153 -0.749*** -0.493*** 3 0.633** -0.362** -0.081 0.152 4 0.648** -0.405** -0.004 -0.004 5 0.180 -0.071 6 0.102 0.018 -0.836** -0.688** 7 7 0.142 x -0.159 x	$\begin{array}{ c c c c c c c c c c c c c c c c c c c$	Baseline OECD 4-year periods Rule dummy 1 -0.647*** -0.404** -0.358** -0.549*** 2 -0.269 -0.153 -0.019 -0.300 -0.749*** -0.493*** -0.639*** -0.396** 3 -0.633** -0.362** -0.346** -0.396** -0.081 -0.152 -0.053 0.126 4 -0.648** -0.405** -0.379** -0.554*** -0.004 -0.004 -0.032 0.049 5 -0.180 -0.071 -0.012 -0.077 6 -0.102 0.018 0.135 0.017 -0.836** -0.688** -0.756*** -0.782** 7 -0.142 x 0.106 0.005 -0.159 x -0.372 -0.186	$ \begin{array}{ c c c c c c c c c } \hline Baseline & OECD & 4-year & Rule & POSET \\ \hline periods & dummy \\ \hline 1 & -0.647^{***} & -0.404^{**} & -0.358^{**} & -0.549^{***} & -1.317^{***} \\ \hline 2 & -0.269 & -0.153 & -0.019 & -0.300 & -0.467 \\ -0.749^{***} & -0.493^{***} & -0.639^{***} & -0.396^{**} & -1.112^{**} \\ \hline -0.749^{***} & -0.362^{**} & -0.346^{**} & -0.586^{***} & -1.301^{***} \\ \hline -0.081 & -0.152 & -0.053 & 0.126 & -0.085 \\ \hline 4 & -0.648^{**} & -0.405^{**} & -0.379^{**} & -0.554^{***} & -1.315^{***} \\ \hline -0.004 & -0.004 & -0.032 & 0.049 & 0.012 \\ \hline 5 & -0.180 & -0.071 & -0.012 & -0.077 & -0.608 \\ \hline 6 & -0.102 & 0.018 & 0.135 & 0.017 & -0.419 \\ \hline -0.836^{**} & -0.688^{**} & -0.756^{***} & -0.782^{**} & -1.387^{*} \\ \hline 7 & -0.142 & x & 0.106 & 0.005 & -0.394 \\ \hline -0.159 & x & -0.372 & -0.186 & -0.588 \\ \hline \end{array}$

Table 6

Results of the robustness analysis

Notes. * indicates significance at a 10% level, ** indicates significance at a 5% level, *** indicates significance at a 1% level. The OECD sample covers 41 countries and four 5-year time windows over the period 1996–2015 (155 observations). Other controls as well as cross-sectional fixed effects according to equation (1) are also included, but not reported. Coefficient t-test results are based on heteroskedasticity robust standard errors.

5.5. The effect of fiscal policy on output volatility

Finally, after documenting the presence of the relationship between fiscal rules and policy volatility, we proceed with assessing whether rules also contribute to macroeconomic stability. We first start by investigating the possibility of a direct relationship between fiscal restrictions and output volatility.

The first two columns of Table 7 report the estimation results when output volatility is directly related to the existence of fiscal rules, without controlling for fiscal volatility. The presence of budget balance rules has a negative effect on output volatility; however its significance is weak. The expenditure rule variable is not significant at any conventional level. Once we control for the conduct of discretionary fiscal policy, by including one of the fiscal volatility measures, the effect of fiscal rules shrinks in magnitude and does not appear statistically significant (see Table 7, columns 3–10).

 Table 7

 Direct effect of fiscal rules on output volatility

					Fisc	al volatilit	y measure	ed as:		
			Prin expen	nary Iditure	Public consumption		Public consumption and investment		Primary balance	
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)
Dep. var.: Output volatility										
Fiscal volatility	XXX	XXX	0.945***	0.971***	0.895***	0.921***	0.940***	0.974***	1.032***	1.032***
Budget balance rule	-0.810* (0.431)		$(0.096) \\ -0.229 \\ (0.249)$	(0.093)	(0.127) -0.179 (0.244)	(0.121)	(0.125) -0.222 (0.249)	(0.120)	(0.120) -0.244 (0.223)	(0.120)
Expenditure rules		-0.335		-0.054		0.069		0.118		-0.153
		(0.309)		(0.182)		(0169)		(0.171)		(0.200)
R-sq within	0.051	0.062	0.593	0.605	0.593	0.604	0.561	0.573	0.689	0.683

Notes. * indicates significance at a 10% level, ** indicates significance at a 5% level, *** indicates significance at a 1% level. The sample covers the EU28 countries and four 5-year time windows over the period 1996–2015 (110 observations). Coefficient t-test results are based on heteroskedasticity robust standard errors. These standard errors are presented in parentheses. Controls as well as cross-sectional fixed effects are also included, but not reported for the sake of brevity.

Next, we relate output volatility to fiscal volatility and use fiscal rules variables as instruments for fiscal volatility as the latter is considered endogenous. As discussed in Fatas and Mihov (2006), fiscal rules might have an indirect effect on macroeconomic volatility, i.e. intermediated by fiscal policy. Alongside fiscal rules variables, we employ other instruments, previously used when estimating equation (1). The estimation results indicate that the standard deviation of discretionary fiscal policy shocks is a positive and statistically significant determinant of output volatility across all specifications (see Table 8). Instrument validity is not rejected by the estimated Hansen J-test and C-test statistics, which is an alternative indication that fiscal rules do not have a direct effect on macroeconomic stability.²⁸ With regard to instrument quality, the null hypothesis of instrument irrelevance is rejected by both the SW first-stage chi-squared and F-test statistics for most specifications except the ones where primary balance is used as a dependent variable.²⁹

To sum up, the estimation results indicate that volatile conduct of discretionary fiscal policy results in higher macroeconomic volatility. Fiscal policy restrictions, such as the presence of national fiscal rules (in particular those designed to be unaffected by the current state of the business cycle), reduce the use of discretionary fiscal policy and could thus indirectly contribute to smoothing out the business cycles.

²⁸ For a description of the instrument validity and relevance tests and the calculation of heteroskedasticity robust standard errors, see Schaffer (2005).

²⁹ For details on these tests, see Sanderson and Windmeijer (2016) who provide a modification and improvement of the method described by Angrist and Pischke (2009).

	Primary e	Primary expenditure		Public consumption		nsumption estment	Primary balance	
Second stage								
Dep. var.: Output volatility								
Fiscal volatility	0.808***	0.604**	0.943***	0.745**	0.944***	0.692**	1.375***	1.267**
	(0.238)	(0.292)	(0.279)	(0.357)	(0.283)	(0.345)	(0.333)	(0.533)
R-sq within	0.581	0.514	0.427	0.571	0.568	0.515	0.668	0.688
First stage								
Dep. var.: Fiscal volatility								
Budget balance rule	-0.517*		-0.647**		-0.560*		-0.471*	
-	(0.269)		(0.290)		(0.285)		(0.264)	
Expenditure rules		-0.082		-0.180		-0.199		0.061
		(0.220)		(0.240)		(0.217)		(0.265)
R-sq within	0.271	0.230	0.200	0.147	0.203	0.152	0.209	0.174
Hansen J-stat	3.135	1.112	2.216	1.477	2.939	1.430	0.245	0.507
C-stat	2.378	0.414	0.685	0.003	1.429	0.030	0.114	0.372
SW chi-sq stat	18.73***	16.80***	13.70***	10.00**	14.56***	13.32***	6.86	3.65
SW F-stat	4.28***	3.84***	3.13**	2.29**	3.33**	3.05**	1.57	0.83

 Table 8

 Indirect effect of fiscal rules on output volatility

Notes. 2SLS estimates of equation (2) are provided in the upper panel, but the first-stage regression results for fiscal volatility – in the lower one. * indicates significance at a 10% level, ** indicates significance at a 5% level, *** indicates significance at a 1% level. The sample covers the EU28 countries and four 5-year time windows over the period 1996–2015 (110 observations). Other controls as well as cross-sectional fixed effects according to equation (2) are also included in the first and second stages, but not reported. Coefficient t-test results are based on heteroskedasticity robust standard errors. These standard errors are presented in parentheses. 2SLS estimates use the fiscal rule, government fragmentation, instruments of government size (the log of total population, the dependency ratio, the urbanisation rate) and other included exogenous variables (log of GDP per capita and openness) as instruments for fiscal volatility. Hansen J-stat is a test of instrument validity (i.e. they are not correlated with the error term of the main equation). The failure to reject the null hypothesis implies that instruments are valid. Similarly, C-stat is a test of validity of a fiscal rule instrument only. The SW first-stage chi-squared and F-stat are tests of instrument relevance. The null of the SW chi-sq test means that fiscal volatility is unidentified, the null of the SW chi F test means that fiscal volatility is weakly identified.

6. CONCLUSIONS

In this study, we analyse whether fiscal rules have a positive stabilising effect beyond their widely recognised contribution to the reduction of the deficit bias. In addition to the existing literature, we also look at the design properties of fiscal rules and compliance with the rules. We employ panel data for the EU28 countries over the period 1996–2015 and estimate the effect of budget balance and expenditure rules on fiscal policy volatility. We examine the impact of rule design and compliance by including interactions with the fiscal rule variable.

Overall, our results confirm the finding of previous studies that the existence of fiscal rules is associated with less volatile discretionary fiscal policy and suggest that strong, properly defined numerical rules act as an anchor for fiscal policy makers. However, in the light of our results this general conclusion has to be interpreted with caution as we confirm it for a specific set of fiscal rules that partial out the current state in the business cycle, i.e. which are "a-cyclical". The basic stabilising effect of such rules is not dampened if they are not complied with year-to-year. However, it does not mean that more stringent rules might not lead to more compliance and in consequence also to stronger effects.

The results of our study contribute to the current debate on the reform of the EU supranational fiscal framework. The existing set of rules stipulated in the SGP has become very complex and opaque, thus reducing their credibility and effectiveness. Several proposals to overhaul the SGP suggest a framework based on a reference value for public debt ratio with an operational annual limit for expenditure growth (European Fiscal Board (2018)). Although an appropriate fiscal rule should mainly be selected due to its ability of counteracting the deficit bias and ensuring the sustainability of public debt, our results suggest that a-cyclical rules could also be an effective tool in achieving more stability in the conduct of fiscal policy. To this end, an expenditure limit should not be related to contemporaneous variables but rather to less time-varying or longer-term variables like potential growth. Furthermore, if one element of the new framework would still be a budget balance rule, it should be in cyclically adjusted or structural terms rather than a nominal limit. Various proposals currently being discussed already take into account both considerations (e.g. Christofzik et al. (2018), Darvas et al. (2018)) such that an accordingly designed new framework might contribute to less volatile discretionary fiscal policy.

It should be kept in mind though that this study is the first attempt to look at design properties of and compliance with fiscal rules in connection with fiscal policy volatility. Thus, it can only be a starting point for future research. A more rigorous analysis of fiscal rule designs might reveal a clearer picture of the properties which make rules more stable and which do not. Furthermore, if rules work as an anchor for politicians and the public, it would be worth investigating how this function can be promoted, e.g. through more transparency, monitoring and accountability of fiscal policy.

7. APPENDIX

Table A1

List of variables used in the study, their definition and source

Variable	Data source	Definition
Output volatility	Ameco	Standard deviation of annual change in real GDP per capita (× 100; within-period average)
Primary expenditure	Eurostat	General government total expenditure excluding interest outlays, deflated using GDP deflator (2010 base year)
Public consumption	Eurostat	Final consumption expenditure of general government (chain-linked volumes; 2010)
Government investment	Eurostat	General government investment expenditure, deflated using gross fixed capital formation deflator (2010 base year)
Primary balance	Eurostat	General government budget balance excluding interest expenditure (% of GDP)
Inflation	IMF	Consumer price index (% change)
Real GDP	Eurostat	Gross domestic product at market prices (chain-linked volumes; 2010)
Real GDP per capita	Eurostat	Gross domestic product at market prices per capita (chain-linked volumes; 2010), within-period average
Openness	Eurostat	Sum of exports and imports as a share of GDP (within-period average)
Government size	Eurostat	Primary expenditure as a share of GDP (within-period average)
Population size	Eurostat	Population on 1 January (within-period average)
Urbanisation rate	World Development Indicators database	Urban population (% of total; within-period average)
Dependency ratio	Eurostat	Old-age dependency ratio (population 65 and over to population 15 to 64 years; %; within-period average)
Government	The Database of	The sum of the squared seat shares of the parties in government
fragmentation	Political Institutions	(within-period average)
Budget balance rule	Badinger and Reuter (2015), Schaechter et al. (2012)	The share of years when a country has a fiscal rule in place; POSET index of stringency (within-period average); IMF index (within-period average)
Expenditure rules	Badinger and Reuter (2015), Schaechter et al. (2012)	The share of years when a country has a fiscal rule in place; POSET index of stringency (within-period average); IMF index (within-period average)
CAB rule	Own calculations based on IMF (2017)	The share of years when a rule was set in cyclically adjusted or structural terms
Expenditure rules linked to potential growth	Own calculations based on IMF (2017)	The share of years when a rule linked expenditure to potential GDP developments
Compliance	Reuter (2017)	The share of years when a rule was complied with

Table A2 **Descriptive statistics**

Variable	Mean	Median	Standard deviation	Min.	Max.
Output volatility	2.420	1.665	2.154	0.385	11.258
Fiscal volatility					
- primary expenditure (GMM)	0.723	0.506	0.569	0.034	2.894
- public consumption (GMM)	0.691	0.537	0.587	0.100	3.206
- public consumption and investment (GMM)	0.616	0.456	0.545	0.099	2.910
- primary balance (GMM)	0.736	0.544	0.548	0.087	2.835
- primary expenditure (2SLS)	0.477	0.401	0.309	0.109	2.606
– public consumption (2SLS)	0.926	0.749	0.760	0.119	4.086
- public consumption and investment (2SLS)	0.773	0.613	0.683	0.100	3.742
- primary balance (2SLS)	0.877	0.628	0.666	0.089	3.447
Log of real GDP per capita	9.796	9.889	0.726	7.979	11.293
Openness	109.474	89.531	60.124	44.966	363.035
Government size	42.135	42.018	5.707	30.370	55.332
Log of population size	6.885	6.937	0.618	5.584	7.916
Urbanisation rate	71.761	69.806	12.066	49.083	97.774
Dependency ratio	23.493	23.740	3.776	15.960	32.560
Government fragmentation	0.805	0.813	0.072	0.539	0.910
Budget balance rule	0.343	0.000	0.428	0.000	1.000
CAB rule	0.257	0.000	0.429	0.000	1.000
Expenditure rules	0.314	0.000	0.433	0.000	1.000
Expenditure rules linked to potential growth	0.098	0.000	0.300	0.000	1.000

Note. Descriptive statistics of fiscal rule related variables is analysed in detail in Section 4.

Table A3

Pairwise correlations among the measures of discretionary fiscal policy volatility, panel (2SLS) estimates

Budget item	Primary expenditure	Public consumption	Public consumption and investment	Primary balance
Primary expenditure	1.000			
Public consumption	0.247	1.000		
Public consumption and investment	0.227	0.947	1.000	
Primary balance	0.442	0.692	0.574	1.000

Note. Authors' calculations based on estimates of equation (3).

Table A4

Pairwise correlations among the measures of discretionary fiscal policy volatility; panel (GMM) estimates

Budget item	Primary expenditure	Public consumption	Public consumption and investment	Primary balance
Primary expenditure	1.000			
Public consumption	0.591	1.000		
Public consumption and investment	0.628	0.961	1.000	
Primary balance	0.717	0.650	0.605	1.000

Note. Authors' calculations based on estimates of equation (3).

Table A5 Effect of fiscal rules on fiscal volatility (fiscal volatility is estimated with 2SLS)

	Primary expenditure		Public consumption		Public consumption and investment		Primary balance	
Dep. var.: Fiscal volatility								
Budget balance rule	-0.281** (0.140)		-0.825** (0.367)		-0.733** (0.357)		-0.587* (0.322)	
Expenditure rules		-0.046 (0.107)		-0.117 (0.319)		-0.128 (0.274)		0.057 (0.319)
R-sq within	0.271	0.198	0.184	0.123	0.155	0.092	0.205	0.168

Notes. * indicates significance at a 10% level, ** indicates significance at a 5% level, *** indicates significance at a 1% level. The sample covers the EU28 countries and four 5-year time windows over the period 1996–2015 (110 observations). Cross-sectional fixed effects are included. Coefficient t-test results are based on heteroskedasticity robust standard errors. These standard errors are presented in parentheses.

Table A6
Effect of fiscal rules on fiscal volatility by type of rule (fiscal volatility is estimated with
2SLS)

	Primary expenditure		Public con	blic consumption		Public consumption and investment		balance
Dep. var.: Fiscal volatility								
Budget balance rule	-0.154		-0.362		-0.404		-0.240	
	(0.145)		(0.373)		(0.399)		(0.347)	
Budget balance rule \times CAB	-0.250**		-0.919***		-0.654**		-0.690**	
rule	(0.109)		(0.346)		(0.313)		(0.317)	
Expenditure rules		-0.039		-0.010		-0.045		0.128
		(0.106)		(0.293)		(0.252)		(0.313)
Expenditure rules × Potential		-0.075		-1.146**		-0.890*		-0.764**
output benchmark		(0.119)		(0.483)		(0.449)		(0.334)
F-test	7.69***	1.37	14.94***	6.11**	14.80***	6.07**	8.51***	3.81*
Joint-sign test	3.31**	1.27	5.40***	3.59**	4.51**	3.33*	3.42**	6.02***
R-sq within	0.262	0.199	0.233	0.170	0.189	0.130	0.238	0.193

Notes. * indicates significance at a 10% level, ** indicates significance at a 5% level, *** indicates significance at a 1% level. The sample covers the EU28 countries and four 5-year time windows over the period 1996–2015 (110 observations). Cross-sectional fixed effects are included. Coefficient t-test results are based on heteroskedasticity robust standard errors. Those standard errors are presented in parentheses. The null of F-test is that the sum of coefficients on a fiscal rule variable and its interaction is equal to zero. The null of joint significance test means that both a coefficient on a fiscal rule variable and on its interaction are equal to zero.

Table A7 Effect of compliance with fiscal rules on fiscal volatility (fiscal volatility is estimated with 2SLS)

	Prima	ry expen	diture	Public consumption		Public consumption and investment			Primary balance			
Dep. var.: Fiscal volatility												
Budget balance	-0.268*	-0.156		-0.831**	-0.325		-0.752**	-0.381		-0.589*	-0.191	
rule	(0.149))	(0.146)		(0.373)	(0.374)		(0.361)	(0.400)		(0.335)	(0.348)	
Expenditure rules			-0.053			-0.040			-0.018			0.087
			(0.104)			(0.310)			(0.243)			(0.343)
Budget balance		-0.249**			-1.017**			-0.715**			-0.816**	¢
rule × CAB rule		(0.122)			(0.392)			(0.352)			(0.351)	
Budget balance	-0.070			0.028			0.101			0.008		
rule × Compliance	(0.179)			(0.478)			(0.346)			(0.518)		
Expenditure rules ×			0.027			-0.317			-0.452			-0.122
Compliance			(0.183)			(0.596)			(0.565)			(0.530)
Budget balance		-0.006			-0.496			0.211			0.433	
rule \times CAB rule \times		(0.159)			(1.969)			(0.390)			(0.510)	
Compliance												
F-test	4.21**	2.28	0.01	4.76**	3.01*	0.31	4.54**	2.39	0.70	1.72	0.70	0.00
Joint-sign test	3.20*	2.18*	0.06	3.75**	3.97**	0.16	3.79**	3.05**	0.35	2.08	2.33*	0.03
R-sq within	0.242	0.263	0.198	0.201	0.257	0.126	0.203	0.240	0.099	0.210	0.247	0.169
Hansen J-stat	2.236	3.755	1.110	2.493	2.229	1.180	3.071	3.477	1.304	1.576	2.234	0.646
C-stat	1.673	0.000	0.057	0.487	0.070	0.003	1.170	0.001	0.097	0.047	0.111	0.082
SW chi-sq stat	15.75***	22.45***	13.47**	13.79**	18.83***	5.92	14.66**	18.71***	7.38	7.16	10.12	3.79
SW F-stat	2.84**	3.33***	2.43**	2.49**	2.79**	1.07	2.65**	2.78**	1.33	1.29	1.50	0.68

Notes. * indicates significance at a 10% level, ** indicates significance at a 5% level, *** indicates significance at a 1% level. The sample covers the EU28 countries and four 5-year time windows over the period 1996–2015 (110 observations). Cross-sectional fixed effects are included. Coefficient t-test results are based on heteroskedasticity robust standard errors. Those standard errors are presented in parentheses. The null of F-test means that the sum of coefficients on a fiscal rule variable and its interaction is equal to zero. The null of joint significance test means that both a coefficient on a fiscal rule variable and on its interaction are equal to zero.

Table A8

Effect of fiscal rules and compliance with them on fiscal volatility measured by volatility in public consumption accounting for all three rules in the single regression (estimated with GMM)

	Pub	lic consumptio	n
	Effect of	Effect of	Effect of
	rules	rule type	compliance
Dep. var.: Fiscal volatility			
Budget balance rule	-0.655**	-0.206	-0.152
	(0.286)	(0.274)	(0.328)
Expenditure rules	0.025	0.035	0.006
	(0.211)	(0.208)	(0.244)
Budget balance rule \times CAB rule		-0.717***	-0.801**
		(0.265)	(0.338)
Expenditure rules × Potential output benchmark		-0.592**	-0.580**
		(0.265)	(0.283)
Budget balance rule × Compliance			-0.145
			(0.649)
Expenditure rules × Compliance			0.097
			(0.431)
Budget balance rule × CAB rule × Compliance			0.285
			(0.705)
R-sq within	0.200	0.276	0.278

Notes. * indicates significance at a 10% level, ** indicates significance at a 5% level, *** indicates significance at a 1% level. The sample covers the EU28 countries and four 5-year time windows over the period 1996–2015 (110 observations). Controls as well as cross-sectional fixed effects are also included, but not reported for the sake of brevity. Coefficient t-test results are based on heteroskedasticity robust standard errors. These standard errors are presented in parentheses.

8. REFERENCES

Acemoglu, Daron, Zilibotti, Fabrizio (1997). Was Prometheus Unbound by Chance? Risk, Diversification, and Growth. *Journal of Political Economy*, vol. 105, issue 4, pp. 709–751.

Afonso, António, Agnello, Luca, Furceri, Davide (2010). Fiscal Policy Responsiveness, Persistence, and Discretion. *Public Choice*, vol. 145, issue 3–4, October 2010, pp. 503–530.

Agnello, Luca, Furceri, Davide, Sousa, Ricardo M. (2013). How Best to Measure Discretionary Fiscal Policy? Assessing Its Impact on Private Spending. *Economic Modelling*, vol. 34, issue C, pp. 15–24.

Alesina, Alberto, Perotti, Roberto, Giavazzi, Francesco, Kollintzas, Tryphon (1995). Fiscal Expansions and Adjustments in OECD Countries. *Economic Policy*, vol. 10, No. 21, October 1995, pp. 205–248.

Angrist, Joshua David, Pischke, Jörn-Steffen (2009). *Mostly Harmless Econometrics: An Empiricist's Companion*. Princeton: Princeton University Press. 392 p.

Arellano, Manuel, Bond, Stephen (1991). Some Tests of Specification for Panel Data: Monte Carlo Evidence and an Application to Employment Equations. *The Review of Economic Studies*, vol. 58, issue 2, April 1991, pp. 277–297.

Arellano, Manuel, Bover, Olympia (1995). Another Look at the Instrumental Variable Estimation of Error-Components Models. *Journal of Econometrics*, vol. 68, issue 1, July 1995, pp. 29–51.

Badinger, Harald (2009). Fiscal Rules, Discretionary Fiscal Policy and Macroeconomic Stability: An Empirical Assessment for OECD Countries. *Applied Economics*, vol. 41, issue 7, pp. 829–847.

Badinger, Harald, Reuter, Wolf Heinrich (2015). Measurement of Fiscal Rules: Introducing the Application of Partially Ordered Set (POSET) Theory. *Journal of Macroeconomics*, vol. 43, issue C, pp. 108–123.

Badinger, Harald, Reuter, Wolf Heinrich (2017). The Case for Fiscal Rules. *Economic Modelling*, vol. 60, issue C, pp. 334–343.

Blanchard, Olivier Jean (1993). Suggestions for a New Set of Fiscal Indicators. *In:* H. A. A. Verbon and F. A. A. M. van Winden (editors), *The New Political Economy of Government Debt*, Elsevier Science Publishers.

Blundell, Richard William, BOND, Stephen (1998). Initial Conditions and Moment Restrictions in Dynamic Panel Data Models. *Journal of Econometrics*, vol. 87, issue 1, August 1998, pp. 115–143.

Christofzik, Désirée I., Feld, Lars P., Reuter, Wolf Heinrich, Yeter, Mustafa (2018). *Uniting European Fiscal Rules: How to Strengthen the Fiscal Framework*. German Council of Economic Experts Working Paper Series 04/2018, September 2018. 28 p.

Cordes, Till, Kinda, Tidiane, Muthoora, Priscilla, Weber, Anke (2015). *Expenditure Rules: Effective Tools for Sound Fiscal Policy*? IMF Working Paper 15/29, February 2015. 28 p.

Darvas, Zsolt, Martin, Philippe, Ragot, Xavier (2018). *European Fiscal Rules Require a Major Overhaul*. Bruegel Policy Contribution Issue No. 18, October 2018. 18 p.

Delgado-Téllez, Mar, Lledó, Victor D., Pérez, Javier J. (2017). On the Determinants of Fiscal Non-Compliance: An Empirical Analysis of Spain's Regions. IMF Working Paper 17/5, January 2017. 33 p.

Debrun, Xavier, Moulin, Laurent, Turrini, Alessandro, Ayuso-I-Casals, Joaquim, Kumar, Manmohan S. (2008). Tied to the Mast? National Fiscal Rules in the European Union. *Economic Policy*, vol. 23, issue 4, pp. 297–362.

Easterly, William, Islam, Roumeen, Stiglitz, Joseph Eugene (2001). *Shaken and Stirred: Explaining Growth Volatility. In:* B. Pleskovic and N. Stern (eds.), Annual World Bank Conference on Development Economics. Washington: The World Bank, pp. 191–211.

EC (2016). Fiscal Rules Database (<u>https://ec.europa.eu/info/business-economy-euro/indicators-statistics/economic-databases/fiscal-governance-eu-member-states/numerical-fiscal-rules-eu-member-countries_en#downloads).</u>

European Fiscal Board (2018). Annual Report 2018. 102 p.

Fatas, Antonio, Mihov, Ilian (2001). Government Size and Automatic Stabilizers: International and Intranational Evidence. *Journal of International Economics*, vol. 55, issue 1, pp. 3–28.

Fatas, Antonio, Mihov, Ilian (2003). The Case for Restricting Fiscal Policy Discretion. *The Quarterly Journal of Economics*, vol. 118, issue 4, pp. 1419–1447.

Fatas, Antonio, Mihov, Ilian (2006). The Macroeconomic Effects of Fiscal Rules in the US States. *Journal of Public Economics*, vol. 90, issue 1–2, pp. 101–117.

Galí, Jordi, Perotti, Roberto, Lane, Philip R., Richter, Wolfram F. (2003). Fiscal Policy and Monetary Integration in Europe. *Economic Policy*, vol. 18, No. 37, pp. 533–572.

Galí, Jordi (1994). Government Size and Macroeconomic Stability. *European Economic Review*, vol. 38, issue 1, pp. 748–756.

Di Giovanni, Julian, Levchenko, Andrei A. (2009). Trade Openness and Volatility. *The Review of Economics and Statistics*, vol. 91, issue 3, August 2009, pp. 558–585.

Greenwood, Jeremy, Jovanovic, Boyan (1990). Financial Development, Growth, and the Distribution of Income. *Journal of Political Economy*, vol. 98, No. 5, Part 1 (October 1990), pp. 1076–1107.

IMF (2009). *Fiscal Rules – Anchoring Expectations for Sustainable Public Finances*. Fiscal Affairs Department, IMF. 72 p.

IMF (2017). Fiscal Rules at a Glance. IMF Background Paper, March 2017. 82 p.

Kopits, George, Symansky, Steven (1998). *Fiscal Policy Rules*. IMF Occasional Paper No. 162, July 1998. 45 p.

Rodrik, Dani (1998). Why Do More Open Economies Have Bigger Governments? *Journal of Political Economy*, vol. 106, issue 5, pp. 997–1032.

Reuter, Wolf Heinrich (2015). National Numerical Fiscal Rules: Not Complied with, but Still Effective? *European Journal of Political Economy*, vol. 39, issue C, pp. 67–81.

Reuter, Wolf Heinrich (2017). *When and Why Do Countries Break Their National Fiscal Rules*? German Council of Economic Experts, Working Paper 01/2017, May 2017. 28 p.

Sacchi, Agnese, Salotti, Simone (2015). The Impact of National Fiscal Rules on the Stabilisation Function of Fiscal Policy. *European Journal of Political Economy*, vol. 37, issue C, pp. 1–20.

Sanderson, Eleanor, Windmeijer, Frank (2016). A Weak Instrument F-Test in Linear IV Models with Multiple Endogenous Variables. *Journal of Econometrics*, vol. 190, issue 2, February 2016, pp. 212–221.

Schaechter, Andrea, Kinda, Tidiane, Budina, Nina, Weber, Anke (2012). Fiscal Rules in Response to the Crisis – Toward the "Next-Generation" Rules. A New Dataset. IMF Working Paper 12/187, July 2012. 48 p.

Schaffer, Mark E. (2005). XTIVREG2: Stata Module to Perform Extended IV/2SLS, GMM and AC/HAC, LIML and k-class Regression for Panel Data Models, Statistical Software Components S456501, Boston College Department of Economics.

Wren-Lewis, Simon (2011). *Comparing the Delegation of Monetary and Fiscal Policy*. University of Oxford, Department of Economics, Discussion Paper No. 540, March 2011. 36 p.